Author’s response to reviews

Title: Gastric MALT lymphoma presented with primary perforation in an adolescent: a case report

Authors:

Yu-Tang Chang (890300@ms.kmuh.org.tw)
Ming-Yii Huang (miyihu@gmail.com)
Hsiang-Hung Shih (hsianghung@gmail.com)
Chun-Chieh Wu (lazzz.wu@gmail.com)
Tzu-Ying Lu (940442@ms.kmuh.org.tw)
Pei-Chin Lin (cooleylin@gmail.com)

Version: 1 Date: 13 May 2018

Author’s response to reviews:

Editor's comments:

1. Thank you for including a CARE checklist alongside your manuscript. However, we ask that you revise the CARE checklist and complete it, as the instructions indicate, by detailing the locations in your manuscript where each checklist item is addressed, as opposed to filling it out with checkmarks.

=> The CARE checklist was revised.

This is so your manuscript can be evaluated fully at peer review.

2. Please rewrite the sentences in the Discussion section beginning with "Long-term clinical outcome of gastric MALT..." and "Surgery no longer plays a role..." as these sections of your manuscript overlap with text in other, published, papers.

=> Those sentences were rewritten.
3. In accordance with BMC submission guidelines for Case Reports, please combine your sections, Discussion and Conclusions, into one cohesive section headed as, “Discussion and Conclusions.”

https://bmcpediatr.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report

=> The sections were combined.

4. In the Authors' contributions section, please refer to all authors by their initials rather than full names.

=> The full names were replaced by initials.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.